Author's response to reviews

Title: Leiomyomatosis peritonealis disseminata in association with Currarino syndrome?

Authors:

Carmine Nappi (nappi@unina.it)
Attilio Di Spiezio Sardo (cdispie@tin.it)
Vincenzo Dario Mandato (dariomandato@virgilio.it)
Giuseppe Bifulco (giuseppebifulco@hotmail.com)
Elisa Merello (m73eli@libero.it)
Antonio Savanelli (savaneelli@unina.it)
Chiara Mignogna (chiaramignogna@tin.it)
Valeria Capra (valeriacapra@ospedale-gaslini.ge.it)
Maurizio Guida (maurizioguida@unina.it)

Version: 6 Date: 20 April 2006

Author's response to reviews: see over
Answer to referees

The Manuscript was revised by the “Manuscript Presentation Service” and the format was changed according to the formatting checklist provided by BMC.

1. Equal contributions- “Please note that we can only allow two authors to be listed as equal contributors. Can we therefore ask you to revise the author details on the title page accordingly”.

   The text has been modified according to the referee’s suggestion (page 10).

2. Abstract- “Please write 'a 27 year-old patient' in the first sentence of the Case presentation section. Please also remove the capitalization from 'Department' in the same section. Please also write 'The three largest abdominal-pelvic masses' in the second paragraph of the same section”.

   The text has been modified according to the referee’s suggestion (page 2).

3. Case presentation- Please revise the first sentence to read 'A 27 year-old Caucasian'.

   The text has been modified according to the referee’s suggestion (page 4).

4. “Please also revise the beginning of the fourth paragraph under the same section to read 'Two months after the birth...’”.

   The text has been modified according to the referee’s suggestion (page 4, fourth sentence in the first paragraph).